Author's response to reviews

Title: 'Reaching the hard to reach' - lessons learned from the VCS (voluntary and community Sector). A qualitative study.

Authors:  
Sarah M Flanagan (s.m.flanagan@bham.ac.uk)  
Beverley Hancock (b.hancock@bham.ac.uk)

Version: 8 Date: 23 March 2010

Author's response to reviews:

Dear Editors,

MS: 1549536349235806

'Reaching the Hard to Reach' - Lessons learned from the VCS (Voluntary and Community Sector) Sarah M Flanagan and Beverley Hancock

Thank you for your comments on the above submission.

I have amended the manuscript accordingly undertaking the major and minor revisions.

Yours sincerely

Sarah Flanagan